Author’s response to reviews

Title: Nationwide citizen access to their health data: Analysing and comparing experiences in Denmark, Estonia and Australia

Authors:

Christian Nøhr (cn@plan.aau.dk)
Liisa Parv (parvliisa@gmail.com)
Pille Kink (Pille.Kink@itk.ee)
Elizabeth Cummings (Elizabeth.Cummings@utas.edu.au)
Helen Almond (Helen.Almond@utas.edu.au)
Jens Rahbek Nørgaard (JRN@medcom.dk)
Paul Turner (Paul.Turner@utas.edu.au)

Version: 2 Date: 27 Jul 2017

Author’s response to reviews:

In the submitted manuscript we have now:

1) re-formatted the textbooks in page 5, 7, and 9 to figures, and renumbered the succeeding figures.

2) Completed all the headings in the declaration section.

3) Clarified issues in the "Ethics and Content to Participate" declaration.

4) Elaborated the "Availability of Data and Material" declaration.